
diff -r 1ea379515c6d -r 358a72d7bf71 handouts/ho10.tex
--- a/handouts/ho10.tex	Mon Jan 24 00:14:02 2022 +0000
+++ b/handouts/ho10.tex	Tue Mar 22 00:36:18 2022 +0000
@@ -584,12 +584,12 @@
 interpreter, though the implementation is admittedly no
 frills.
 
-\begin{figure}[t]
-\small
-\lstinputlisting[language=Scala]{../progs/inter.scala}
-\caption{The entire code of the interpreter for our
-idealised programming language.\label{code}}
-\end{figure}
+%\begin{figure}[t]
+%\small
+%\lstinputlisting[language=Scala]{../progs/inter.scala}
+%\caption{The entire code of the interpreter for our
+%idealised programming language.\label{code}}
+%\end{figure}
 
 
 \subsubsection*{Static Analysis}




